**Author's response to reviews**

**Title:** Association between Leukocyte Telomere Shortening and Exposure to Traffic Pollution: a Cross-Sectional Study on Traffic Officers and Indoor Office Workers

**Authors:**

Mirjam Hoxha (mirjam.hoxha@unimi.it)  
Laura Dioni (laura.dioni@unimi.it)  
Matteo Bonzini (matteo.bonzini@infinito.it)  
Angela C Pesatori (angela.pesatori@unimi.it)  
Silvia Fustinoni (silvia.fustinoni@unimi.it)  
Domenico Cavallo (domenico.cavallo@unimi.it)  
Michele Carugno (michele.carugno@gmail.com)  
Benedetta Albetti (benedetta.albetti@unimi.it)  
Barbara Marinelli (barbara.marinelli@unimi.it)  
Joel Schwartz (joel@hsph.harvard.edu)  
Pier A Bertazzi (pieralberto.bertazzi@unimi.it)  
Andrea Baccarelli (andrea.baccarelli@unimi.it)

**Version:** 3  **Date:** 2 September 2009

**Author's response to reviews:** see over
Dear Editor,

We are submitting a revised version of the manuscript MS: 1292946641254282 “Association between Leukocyte Telomere Shortening and Exposure to Traffic Pollution: a Cross-Sectional Study on Traffic Officers and Indoor Office Workers”.

We are very thankful for the reviewer’s appreciation of our first revised version. We have now removed bolding, the italics and the pagination throughout the manuscript. As requested, on the title page, the emails are listed in sentence format separating the pairs with semi-colons. We have changed the headings as indicated by email, and on page 7 removed extra spacing. In the Competing Interests “of” has been removed. The references have been properly formatted following the article sample provided on your web site. The tables are presented in closed cell format with all the horizontal and vertical lines visible.

Thanks in advance for your consideration.

Best Regards,

Andrea Baccarelli, MD PhD
Center of Molecular and Genetic Epidemiology
Dept. of Environmental and Occupational Health
University of Milan and Maggiore Hospital,
Mangiagalli and Regina Elena Foundation
Via San Barnaba 8
20122 Milan, ITALY
Tel +39 02 503 20145
Fax: +39 02 503 20103
email: andrea.baccarelli@unimi.it